

Volume 13 • 2025



Research article

Assessing the impact of climate change and a water management programme on white sturgeon physiology in the Nechako River, British Columbia

Muhammed A. Oyinlola^{1,2,*}, Mostafa Khorsandi¹, Rachael Penman³, Madison L. Earhart³, Richard Arsenault⁴, Steve McAdam⁵, Colin J. Brauner³, and André St-Hilaire¹

¹Centre Eau Terre Environnement, Institut National de la Recherche Scientifique, Canadian Rivers Institute, 490 Couronne St, Québec City, QC G1K 9A9, Canada

²Institute for the Oceans and Fisheries, University of British Columbia, 2202 Main Mall, Vancouver, BC V6T 1Z4, Canada

³Department of Zoology, University of British Columbia, 6270 University Boulevard, Vancouver, BC V6T 1Z4, Canada

⁴Hydrology, Climate and Climate Change Laboratory, École de technologie supérieure (ÉTS), 1100 Notre-Dame St W, Montréal, QC H3C 1K3, Canada

⁵Applied Freshwater Ecology Research Unit, BC Ministry of Water Land and Resource Stewardship, University of British Columbia, 2202 Main Mall, Vancouver, BC V6T 1Z4, Canada

*Corresponding author: Institute for the Oceans and Fisheries, University of British Columbia, 2202 Main Mall, Vancouver, BC V6T 1Z4, Canada. Email: m.oyinlola@oceans.ubc.ca

Climate change is impacting river ecosystems, underlining the need for water management strategies to protect native species within these ecosystems. Here, we evaluate the impact of climate change and water management on the physiology of white sturgeon (*Acipenser transmontanus*) in the Nechako River, British Columbia (Canada). Using the CEQUEAU hydrological–thermal model, we simulated daily water temperatures from 1980 to 2099 under two climate scenarios (SSP2-4.5 and SSP5-8.5). We assessed thermal exposure risk (T_e) for different developmental stages of white sturgeon, focusing on the warmest 6-month period. Our findings show that embryos and yolk-sac larvae exhibit resilience, with T_e values consistently <1 under both scenarios, signifying low thermal stress. In contrast, feeding larvae and juveniles experience elevated T_e values, indicating significant future thermal stress. For feeding larvae, T_e values exceeded 1 under both scenarios, reaching up to 1.5 by the mid-century (2050s) and up to 1.8 by the end of the century (2090s) under SSP5-8.5. Juvenile white sturgeon also faced increased thermal risks, with T_e values rising >1 during July and August, reaching 1.4 and 1.8 by the 2050s and 1.8 and 2.0 by the 2090s under SSP5-8.5, compared to the 1980s. These results underscore the need to evaluate the existing water management programme to better accommodate the projected changes in thermal conditions associated with climate change. Additionally, regulated river discharge, which can both increase and decrease downstream temperatures, offers a strategic opportunity to mitigate some climate impacts through strategic dam discharge management.

Lay Summary

Climate change threatens river ecosystems, impacting the Nechako River white sturgeon. Our study shows that sturgeon embryos and yolk-sac larvae exhibit resilience to increasing temperatures, while feeding larvae and juveniles face increased thermal stress. Alternatives to water management, including comprehensive regulation of dam discharges, could help mitigate these climate-driven impacts.

© The Author(s) 2025. Published by Oxford University Press and the Society for Experimental Biology. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/ 1 by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Key words: Climate change, conservation, hydrothermal impact, Nechako River, thermal exposure risk, white sturgeon

Editor: Frank Seebacher

Received 26 October 2024; Revised 27 January 2025; Editorial Decision 4 February 2025; Accepted 12 February 2025

Cite as: Oyinlola MA, Khorsandi M, Penman R, Earhart ML, Arsenault R, McAdam S, Brauner CJ, St-Hilaire A (2025) Assessing the impact of climate change and a water management programme on white sturgeon physiology in the Nechako River, British Columbia. *Conserv Physiol* 13(1): coaf014; doi:10.1093/conphys/coaf014.

.....

.....

Introduction

Climate change is altering riverine ecosystems, emphasizing the need for water management to ensure the health and survival of endemic aquatic species. River water management plays an important role in regulating water quality, quantity and the overall health of aquatic ecosystems, which are affected by changes in temperature, pH, dissolved oxygen, nutrient levels and pollutant concentrations (Carron and Rajaram, 2001; Richter et al., 2003; Song et al., 2019; Zarri et al., 2019; Bernhardt et al., 2022). Effective water management strategies protect and restore aquatic habitats, such as riparian zones, wetlands, and other ecosystems, ensuring that they continue to support endemic species (Richter et al., 2003). These management strategies often regulate discharge in river systems to balance societal demands with the ecological needs of endemic species. Adapting water management plans to address the evolving impacts of climate change has the potential to enhance the resilience and sustainability of aquatic ecosystems.

Water temperature has profound effects on aquatic animals across levels of biological organization, from chemical reactions and metabolism to species distributions (Guderley, 2004; Clarke and Pörtner, 2010; Halsey et al., 2015; Eliason and Anttila, 2017). Changes in water temperature directly impact an organism's overall physiological performance, behaviour and survival. Water temperature impacts chemical reactions and reactions kinetics, which are fundamental to physiological processes. Processes such as digestion and respiration are influenced by water temperature up to an optimal level, beyond which their efficiency declines, impacting the rates of these processes and organismal health (Sardella et al., 2004; Schulte, 2015; Muñiz-González and Martínez-Guitarte, 2020). The effects of water temperature on an individual's physiology can have broader implications for population dynamics and ecosystem interactions. Interactions between water temperature and other environmental factors can either favour or impede organismal reproduction and growth, ultimately shaping the distribution and abundance of species (Perry et al., 2005; Fernandes et al., 2020).

The Nechako white sturgeon (*Acipenser transmontanus*) is endangered due to recruitment failure resulting from the effects of river regulation on critical spawning habitats (DFO, 2014; McAdam *et al.*, 2005). Persistent recruitment failure since 1967 will cause the adult population, currently esti-

mated at 553 individuals, to decline to 200 fish within 30-64 years (van Poorten et al., in review), with hatchery inputs serving as a temporary measure to prevent extirpation. White sturgeon have specific thermal requirements that influence their growth, metabolism and reproductive success (Jay et al., 2020; Penman et al., 2023). Elevated water temperatures, driven by reduced water discharge due to regulation and climate change, can disrupt these physiological processes (Earhart et al., 2023) potentially leading to recruitment failure. Consequently, recruitment challenges may also result from habitat degradation caused by substrate alterations resulting from changes in river discharge (Boucher et al., 2014; McAdam, 2012). Ongoing and future increases in water temperature pose a substantial threat to the recovery and long-term sustainability of white sturgeon populations. Understanding these thermal impacts is essential for developing effective conservation and management strategies to ensure the resilience of white sturgeon in a changing climate.

The Nechako River system in central British Columbia, Canada (Fig. 1A), presents a unique case study for evaluating the effects of climate change and water management on riverine ecosystems. This river is characterized by regulated discharge, significant out-of-basin water diversion, and its location in a northern region, making it particularly susceptible to climate change impacts. Northern regions are experiencing warming at a faster rate than the global average due to Arctic amplification, compounded by extreme weather events that result in prolonged droughts and heavy precipitation (Bush, 2022; Rantanen et al., 2022). The river supports endangered species, including sockeye salmon (Oncorhynchus nerka) and white sturgeon, highlighting the importance of targeted water management to ensure species survival. One notable water management plan is the Summer Temperature Management Program (STMP), implemented in the early 1980s to regulate water releases from the Kenney Dam to maintain downstream river temperatures <20°C during the sockeye salmon spawning migration period from 20 July to 20 August (Macdonald et al., 2012). Although this programme has effectively prevented water temperature from exceeding 20°C at the Vanderhoof station (Macdonald et al., 2012), its focus on sockeye salmon highlights the need for modifications to address the conservation of other species that may differ in their thermal sensitivities.

Recent studies (Earhart et al., 2023; Khorsandi et al., 2023; Oyinlola et al., 2023, 2024; Gatien et al., 2024) empha-



Figure 1: A) Nechako watershed showing the Skins Lake spillway and hydrological stations. B) A schematic diagram of the framework adopted from Oyinlola *et al.* (2023) used in this study. C) Thermal Performance Curve (TPC) showing the relationship between water temperature and fish performance. The blue dashed line (first vertical line) represents the optimal temperature where the performance is at its peak (Te = 1). Thermal exposure risk (Te), refers to the likelihood of adverse effects due to water temperature extremes, with Te values > 1 indicating greater risk and worse performance (Te > 1). The green dashed line (second vertical line) indicates a moderate Te where performance begins to decline. The yellow dashed line (third vertical line) represents higher Te, resulting in a significant decrease in performance (Te > 2). The performance of the fish declines at temperatures lower and higher than the optimal, reflecting the fish's reduced efficiency and increased thermal stress.

size the need to ensure that water temperature management programmes in regulated rivers can address the needs of diverse species and life stages, particularly under changing climatic conditions. Continuous monitoring and analysis of these programmes, alongside the assessment of their effectiveness for other migratory and resident species, is essential. This approach ensures the programme continues to accommodate shifting ecological demands and water resource requirements, ultimately contributing to the resilience and sustainability of the river ecosystem including the Nechako River.

In this study, we employed an established framework (Oyinlola *et al.*, 2023) (Fig. 1B) to assess the impact of climate change on the white sturgeon in the Nechako River. We used a hydrological and river temperature model called Centre Québécois des Sciences de l'Eau (CEQUEAU) to simulate daily water temperatures from 1980 to 2099. Using physiological data from the literature for different developmental stages of white sturgeon, we developed a thermal exposure index. We projected the thermal exposure risk for the warmest 6-month period each year, including the water release management period, under two climate change scenarios (SSP2-4.5 and SSP5-8.5) by the middle and end of the century.

Materials and Methods

We employed a combined modelling framework (Oyinlola et al., 2023) to evaluate the potential impact of climate change on the white sturgeon in the Nechako River. This framework integrates a hydrological model with physiological data specific to each life stage to assess thermal exposure risk (T_e) , which represents the likelihood of adverse effects on the physiological performance and survival due to water temperature extremes, with *Te* values >1 indicating greater risk and worse physiological performance (Fig. 1C) (Oyinlola et al., 2023). We hypothesized that the T_e of the white sturgeon would rise above the optimal value of '1' under climate change. First, we used CEQUEAU to simulate the Nechako River's daily historical water temperature from 1980 to 2019. Next, we developed the thermal exposure index, a quantitative measure of water temperature exposure over time, using white sturgeon physiological data. Last, we projected the T_e for the warmest 6-month period of the year (May-October), which includes the water release management period (20 July-20 August), under two contrasting climate change and socio-economic scenarios: SSP2-4.5 and SSP5-8.5 by mid (the 2050s) and end of the century (the 2090s).

CEQUEAU model

CEQUEAU is a semi-distributed hydrological-thermal model used for simulation and forecasting flow and water temperature (Morin and Couillard, 1990; Morin and Paquet, 1995) (see supplementary information for details). It employs a two-step discretization to spatialize hydrological processes within the watershed area. Each grid square (called CE) is subdivided into a maximum of four sub-areas (polygons called CPs), which act as a hydrological response unit and for which vertical routing and water storage are conceptually represented through three interconnected reservoirs.

In implementing CEQUEAU, parameter calibration is a critical step. This calibration process involves two steps: first, calibrating the hydrological module using observed stream-flow data from hydrometric stations along the Nechako River, and then calibrating the thermal module using water temperature gauges located between the Kenney Dam and Vanderhoof. Manual calibration was initially performed to define the parameter domain, followed by the implementation of an automatic calibration algorithm. The covariance matrix adaptation evolution strategy (CMA-ES) is used for this purpose. CMA-ES has been extensively compared to other optimization algorithms for model calibration in hydrology, consistently demonstrating superior performance in finding global optima and achieving faster convergence (Arsenault *et al.*, 2014).

Thermal tolerance limits and critical habitat for Nechako white sturgeon

We used data from our previous studies on white sturgeon thermal tolerance during each early life stage (i.e. embryo, yolk-sac larvae, feeding larvae and juvenile) (Earhart *et al.*, 2023; Penman *et al.*, 2023). However, for the feeding larvae stage, physiological data and literature were limited, highlighting an urgent need for further studies to fill this data gap. Hence, we relied on expert opinion to define the thermal limits and evaluated scenarios assuming a threshold similar to either the preceding life stage (14–18°C) or the subsequent life stage (14–20°C). Expert opinion was obtained through consensus among co-authors with relevant expertise in sturgeon physiology.

We defined the Optimal Temperature Range (T_{optR}), Suboptimal Temperature Range (ST_{optR}) and Critical Thermal Limit Range (CT_{LR}) for each early life stage (embryo, yolksac larvae, feeding larvae and juvenile) based on the laboratory findings and expert opinion (Table 1). T_{optR} refers to the temperature range in which the fish's physiological performance is at its peak, while ST_{optR} represents the range where some critical functions are lost, but <25% mortality is observed. CT_{LR} is the range where >50% mortality occurs. We calculated the T_e using T_{optR} , ST_{optR} and CT_{LR} .

$$T_{ei} = 0, if [T_a, T_b] < T_{optR}$$

$$\tag{1}$$

$$T_{ei} = 1, if [T_a, T_b] \subset T_{optR}$$

$$\tag{2}$$

$$T_{ei} = 2, if \ [T_a, T_b] \subset ST_{optR} \tag{3}$$

$$T_{ei} = 3, if \ [T_a, T_b] \subset CT_{LR} \tag{4}$$

where T_{ei} is the thermal exposure risk for cell i; T_a and T_b are the minimum and maximum temperature ranges, respectively.

Table 1: White sturgeon thermal exposure risk (T_e) applied for this study, where a value >1 indicates an elevated T_e beyond optimal conditions

Life stage	Variable/- Parameter	Description	Temperature range (° C)	Thermal exposure risk (T _e)	Reference
Embryo		Temperature below optimal temperature (growth/general health condition)	<14	0	(Earhart <i>et al.</i> , 2023).
	T _{optR}	The optimal temperature (growth/general health condition)	14–18	1	(Earhart <i>et al.</i> , 2023).
	ST _{optR}	Sub-optimal temperature (loss of some critical function and $<\!25\%$ mortality)	>18	2	(Earhart <i>et al.</i> , 2023)
	CT _{LR}	Critical temperature (total loss of critical function and >50% mortality)	NA	NA	
Yolk-sac larvae		Temperature below optimal temperature (growth/general health condition)	<14	0	(Earhart <i>et al.</i> , 2023)
	T _{optR}	The optimal temperature (growth/general health condition)	14–20	1	(Earhart et al., 2023)
	ST _{optR}	Sub-optimal temperature (loss of some critical function and $<\!25\%$ mortality)	>20	2	(Earhart <i>et al.</i> , 2023)
	CT _{LR}	Critical temperature (total loss of critical function and >50% mortality)	NA	NA	
Feeding larvae (18°C)		Temperature below optimal temperature (growth/general health condition)	<14	0	Expert opinion
	T _{optR}	The optimal temperature (growth/general health condition)	14–18	1	Expert opinion
	ST _{optR}	Sub-optimal temperature (loss of some critical function and $<\!25\%$ mortality)	>18	2	Expert opinion
	CT _{LR}	Critical temperature (total loss of critical function and >50% mortality)	NA	NA	
Feeding larvae (20° C)		Temperature below optimal temperature (growth/general health condition)	<14	0	Expert opinion
	T _{optR}	The optimal temperature (growth/general health condition)	14–20	1	Expert opinion
	ST _{optR}	Sub-optimal temperature (loss of some critical function and ${<}25\%$ mortality)	>20	2	Expert opinion
	CT _{LR}	Critical temperature (total loss of critical function and $>50\%$ mortality)	NA	NA	
Juvenile		Temperature below optimal temperature (growth/general health condition)	<14	0	
	T _{optR}	The optimal temperature (growth/general health condition)	14–18	-	(Wang <i>et al.</i> , 1985, Wang <i>et al.</i> 1987, DFO, 2014, Cheung, 2019)
	ST _{optR}	Sub-optimal temperature (loss of some critical function and ${<}25\%$ mortality)	>18	2	(Wang <i>et al.</i> , 1985, Wang <i>et al.</i> 1987, Hildebrand <i>et al.</i> 2016)
	CT _{LR}	Critical temperature (total loss of critical function and $>50\%$ mortality)	NA	NA	



Figure 2: Comparative analysis between the simulated water temperature data from the CEQUEAU model and historical water temperature records sourced from the Canadian Climate Archive (https://climate.weather.gc.ca/historical_data/search_historic_data_e.html) for four stations along the Nechako River. (A) Time series of CEQUEAU-simulated water temperatures and historical observed water temperature data for the Vanderhoof station, 2010–19. (B) Time series of CEQUEAU-predicted water temperatures and historical observed water temperature data for the Nautley station, 2010–19. (C) Focused plot for the Vanderhoof station, May–October 2017. (D) Focused plot for the Nautley station, May–October 2017. (E) Focused plot for the Fort Fraser station, May–October 2017. (E) Focused plot for the Fort Fraser 13 s station, May–October 2017.

Climate and socio-economic scenarios

We projected the future changes in Nechako River water temperature using data from eight General Circulation Models (GCMs) as part of the Coupled Model Intercomparison Project Phase 6 (CMIP6) (Eyring *et al.*, 2016) (Supplementary Table S1). We examined two sets of combined climate and socio-economic scenarios. The first scenario, SSP2-4.5, represents an intermediate pathway where the world follows historical social, economic and technological trends. The second scenario, SSP5-8.5, describes a high-emission trajectory characterized by rapid economic growth, reliance on competitive markets, innovation and participatory societies (Riahi *et al.*, 2017). By using these GCMs and scenario combinations, we aimed to capture a range of possible future outcomes for the Nechako River water temperature. The incorporation of multiple models and scenarios enabled a comprehensive assessment of uncertainties associated with climate projections.

Research article

Analysis

We analysed the modelled mean water temperature across the Nechako watershed from 1980 to 2099. We focused on the sections of the river where white sturgeon are distributed. Further, our analyses prioritized the six hottest water temperature months in the Nechako watershed (May– October). We then concentrated on the period when each life stage is present within the Nechako River and analysed the life stage-specific spatio-temporal pattern of T_e for the Nechako River for the embryo, 15 May–24 June; yolk-sac larvae, 24 May–7 July; feeding larvae, 20 June–31 July; and juvenile, 1 May–31 October.

We estimated the average water temperature and T_e in each $0.005 \times 0.005^{\circ}$ C cell calculated from simulated daily water temperature data. To account for variations among climate models, we determined the average T_e across all GCMs and assigned a numerical value to each T_e category ($T_e 0 = 0, T_e$ 1 = 1, T_e 2 = 2, T_e 3 = 3). We calculated the average T_e for the historical period (1980s, averaging 1980-89) and future periods-mid-century (2050s, averaging 2050-59) and endcentury (2090s, averaging 2090-99)-under two emission scenarios: SSP2-4.5 and SSP5-8.5. Similar to the previous study (Oyinlola et al., 2023), we also analysed the average T_e in white sturgeon critical habitats in the Nechako River watershed as identified under Canada's Species at Risk Act (SARA) (Fisheries and Canada, 2014; SARA, 2002). Finally, we analysed the T_e for white sturgeon's early life stages (embryo, yolk-sac larvae, feeding larvae and young-of-theyear juveniles) for the period when each life stage is known to exist in the Nechako River. For the feeding larvae stage, we analysed the two variations in optimal temperature ranges denoted here as feeding larvae18 (optimal range of 14-18°C) and feeding larvae20 (optimal range of 14–20°C).

Results

CEQUEAU model evaluation

We evaluated our water temperature model results using the Root Mean Square Error (RMSE). RMSE is a metric that quantifies the average magnitude of discrepancies between predicted and actual values. We tested predicted against the actual values from four stations along the Nechako River, Vanderhoof, Nautley, Fort Fraser and Fort Fraser 13 s (Fig. 2). The RMSE values for these stations are as follows: 1.27 (Vanderhoof), 1.68 (Nautley), 2.41 (Fort Fraser) and 2.43°C (Fort Fraser 13 s).

Climate trends in the Nechako River

Our results show that the average water temperature across the Nechako River within the 6-month evaluation period ranged from 8.2 to 17°C in historical years—1980s (average from 1980 to 1989) with the lowest water temperature in October and the highest in August (Fig. 3A and B). However, under climate change scenarios SSP2-4.5, by the mid-

.....

century (the 2050s, average from 2050 to 2059), the Nechako River temperature was projected to increase to $18.6^{\circ}C \pm 1.2$ (mean \pm standard deviation (SD)) in the historical hottest month of August, indicating an increase of 9.3% relative to 1980s, while under SSP5-8.5, we projected an increase of 16% (to $19.7^{\circ}C \pm 1.4$). Our results also show that the largest increase in water temperature would occur in September with an increase of 15% (to $15.5^{\circ}C \pm 0.3$) and 19% (to $16.2^{\circ}C \pm 0.3$) under SSP2-4.5 and SSP5-8.5, respectively (Fig. 3A).

By the end of the century (2090s, average from 2090 to 2099) under the SSP2-4.5 scenarios, our results show that August will remain the hottest month in the Nechako River, with the projected increase in water temperature to $18.9^{\circ}C \pm 3.3$, an 11.4% increase relative to the 1980s. However, under SSP5-8.5 in the same time frame, the water temperature is projected to increase to $22.7^{\circ}C \pm 1.5$, a 33% increase (Fig. 3B). Furthermore, under SSP2-4.5, September water temperature would increase by a minimal 5% while October water temperature would decrease by 4% relative to the 1980s. However, our results indicate that the Nechako River temperature in September and October would increase considerably by 42%, reaching 11.62 and 19.3° C, respectively, under SSP5-8.5.

Changes in thermal exposure risk (T_e) of Nechako white sturgeon by the mid and end of the century relative to the 1980s

Changes in thermal exposure risk (T_e) across the Nechako River

Overall, our findings show that future T_e will consistently remain <1 for periods when each embryo and yolk-sac larvae are present in the Nechako River (i.e. 15 May–24 June and 24 May–7 July for embryo and yolk-sac larvae, respectively) under SSP2-4.5 and SSP5-8.5 by the 2050s (Fig. 4A). These results are comparable to historical T_e values estimated for the 1980s. In contrast, both assumed thermal risk ranges for feeding larvae (feeding larvae18 and feeding larvae20), led to T_e values >1 under both scenarios by the 2050s (Fig. 4A). Similar to the 2050s, we projected that the embryo and yolksac larvae T_e by the 2090s will be <1 under both scenarios considered in this study (i.e. SSP2-4.5 and SSP5-8.5) (Fig. 4B). However, for the feeding larvae life stage, we projected T_e values >1 under both scenarios similar to historical T_e values estimated for the 1980s (Fig. 4B).

For the juvenile life stage, T_e , was predicted to exceed 1 during July (1.2) and August (1.6) under SSP2-4.5 by the 2050s relative to the 1980s (Fig. 4C). Under SSP5-8.5 in the same time frame, T_e values were projected to be >1 in July (1.4) and August (1.8) compared to the 1980s. However, by the 2090s compared to the 1980s, the juvenile life stage, T_e , was projected to rise >1 in July (1.4) and August (1.5) under SSP2-4.5 (Fig. 4D). Under SSP5-8.5, we projected T_e


Figure 3: Nechako River temperature during the months examined in this study in the 1980s (average between 1980 and 1989) and under climatic–socioeconomic scenarios; SSP2–4.5 and SSP5–8.5 by A) the mid-century (2050s, average between 2050 and 2059) and B) end-century (average between 2090 and 2099). Percentage change in water temperature relative to the 1980s under both scenarios is indicated on the y-axis as a relative indicator of the water temperature change. The dotted line indicates the recommended water temperature threshold of 20°C.

to increase >1 in June (1.1), July (1.8), August (2.0) and September (1.6) by the 2090s compared to the 1980s.

Changes in thermal exposure risk (T_e) in critical habitats

Our analysis shows that the projected T_e values would remain <1 by the 2050s under both SSP2-4.5 and SSP5-8.5 for embryos and yolk-sac larvae in the Vanderhoof Reach (Figs 5A and 6A-D). For the feeding larvae life stage, irrespective of the optimal temperature variations considered, the T_e values were projected to exceed 1 in both climatic scenarios, although feeding larvae20 under SSP2-4.5 scenarios would only marginally exceed 1 by the 2050s (Figs 5A and 6E-H). The juvenile life stage utilizes a broader range of habitats than the other life stages (Fig. 1C). Our projections show that by the 2050s compared to the 1980s when the juvenile T_e value was only <1 in Fraser Lake, the T_e levels would remain consistently above a value of 1 in all critical habitats during July and August, irrespective of the scenario (SSP2-4.5 and SSP5-8.5) (Fig. 5B). In addition, under the SSP5-8.5, our results show that T_e would be >1 at Keilor's Point and Powerline sections in June.

By the 2090s, our analysis shows that the projected T_e values under the SSP2-4.5, for embryo and yolk-sac larvae in the Vanderhoof Reach, were projected to remain <1(Fig. 5C). In contrast, under SSP5-8.5, the T_e values were projected to be slightly >1 for both life stages. Moreover, within the Vanderhoof Reach, T_e was projected to be >1 for both feeding larvae scenarios under SSP2-4.5 and SSP5-8.5, respectively, although only slightly >1 under SSP2-4.5 scenario by the 2050s (Fig. 5C). As for the juvenile life stage, T_e levels were projected to consistently exceed 1 in all critical habitats during July and August, irrespective of the SSP scenario (SSP2-4.5 and SSP5-8.5) (Fig. 5D, Fig. 7). Specifically, under SSP2-4.5, we projected T_e values >1 for all critical habitats used by the juvenile life stage, except for Fraser Lake, in June. In July and August, all habitats were projected to have T_e values well surpassing the optimal threshold, ranging from 1.78 in Fraser Lake to 1.91 in Powerline. Under SSP 5-8.5, habitats were projected to exhibit T_e values exceeding 1 in June, July, August and September, with the lowest value of 1.50 in Fraser Lake (September) and the highest value of 2.0 observed in August across all habitats.



Figure 4: The thermal exposure risk of white sturgeon embryo, yolk-sac larvae, feeding larvae and juvenile life stages for the period when the life stages are present in the Nechako River under SSP2–4.5 and SSP5–8.5. For embryo, yolk-sac larvae and feeding larvae (larvae18 and larvae20) life stages by the A) mid-century (2050, average between 2050 and 2059) B) end of the century (2099s, average between 2090 and 2099) relative to the 1980s (average between 1980 and 1989). For juvenile life stage by the C) mid-century (2090s, average between 2050 and 2059) D) end of the century (2099s, average between 2090 and 2099) relative to the 1980s (average between 1980 and 1989) for May–October. Error bars represent SD.

Discussion

A key goal of water management is to balance societal needs with ecological diversity in freshwater ecosystems (Richter *et al.*, 2003; Geist, 2011; Mishra *et al.*, 2023). As climate change threatens these systems (Reid *et al.*, 2019; Rose *et al.*,

.....

2023), understanding the interdependence between water management and climate impacts is important for informed decision-making and promoting ecological resilience. This study used an established framework (Oyinlola *et al.*, 2023) that combines a hydrological model and physiological data to evaluate the hydrothermal impact on different white sturgeon life stages under combined climatic and socio-economic scenarios. Results show that river temperatures in the Nechako are expected to rise significantly under SSP2-4.5 and SSP5-8.5 scenarios, especially in July, August and September, thereby increasing thermal exposure risks, particularly for the juvenile life stage.

The CEQUEAU hydrological-thermal model was selected for our study due to its effectiveness in predicting distributed water temperatures at the watershed scale (Khorsandi et al., 2022) and its strong performance in simulating dam release operations and river thermal modelling (St-Hilaire et al., 2015). The lower RMSE values at the Vanderhoof and Nautlev stations indicate the model's accuracy, although it tends to underestimate measured water temperatures. In contrast, the higher RMSE values at the Fort Fraser and Fort Fraser 13 s stations may be attributed to the complexities of riverlake transition such as thermal stratification and localized influences (Daigle et al., 2009; Khorsandi et al., 2022; Auffray et al., 2023). Large nearby water bodies can also affect air temperature and humidity, further affecting model accuracy. These specific characteristics should be considered when applying the CEQUEAU model.

Our study found a consistent rise in the average water temperature of the Nechako River, expected to continue throughout the century under two climate change scenarios. Notable increases are projected for August (9 and 11% by the 2050s and 2090s) and September (16 and 33%) (Fig. 3). This warming trend aligns with findings in BC (Green *et al.*, 2014; Schnorbus *et al.*, 2014) and globally due to climate change (IPCC, 2023). Water temperature increases are expected to have adverse impacts on aquatic ecosystems and their biodiversity (Poff *et al.*, 2002; Palmer *et al.*, 2009; Cheung and Oyinlola, 2019). Our study highlights the importance of integrating species-specific details to enable modelling that provides time, life stage and location-specific evaluations.

Our study provides valuable insights into thermal stress faced by white sturgeon at different life stages and locations in the Nechako River. A T_e value of 1 indicates an optimal temperature for white sturgeon, where their performance is at its peak (Fig. 1B). As T_e values rise >1, there is an increased risk of thermal exposure, with higher T_e values indicating more severe stress and reduced performance (Oyinlola *et al.*, 2023). In the 1980s, white sturgeon across all examined life stages experienced less thermal stress compared to projected conditions under climate change scenarios, as indicated by lower T_e values except for juveniles where T_e was projected to reach 1.14 in August. While the STMP aims to maintain river temperatures <20°C, the temperature management is restricted to periods determined by the presence of migrating

2050s A) Embryo Yolk-sac larvae 2.0 1.5 , 1.0 : 1.0 Thermal exposure ri 0.0 2.0 1.5 1.0 0.5 Feeding larvae 18 Feeding larvae 20 1.0 0.5 0.0 1980s SSP2-4.5 SSP5-8.5 1980s SSP2-4.5 SSP5-8.5 B) Culvert Hole Finmoore 2.0 1.5 1.0 0.5 0.0 Fraser Lake Keilor's Point Thermal exposure risk 0.0110 0.0110 0.00000 0.00100 Γ' Leduc Creek Confluence Powerline Sinkut River Confluence 222 2.0 1.5 1.0 0.5 0.0 May 🗾 July September Month June August October Juvenile 2090s C) Embryo Yolk-sac larvae 2.0 1.5 al exposure risk 0.0 2.0 Feeding larvae 18 Feeding larvae 20 Thermal 1.5 1.0 0.5 0.0 1980s SSP2-4.5 SSP5-8.5 1980s SSP2-4.5 SSP5-8.5 D) Culvert Hole Finmoore 2.0 1.5 1.0 0.5 0.0 Keilor's Point Fraser Lake Thermal exposure risk 000110 000110 000000 000100 Leduc Creek Confluence Powerline SSP2 Sinkut River Confluence 2.0 May 📃 July September Month June August October SSP2 1980 Juvenile

Figure 5: The white sturgeon of the Nechako River early life stages (embryo, yolk-sac larvae, feeding larvae and juvenile) critical habitats thermal exposure risk for the 1980s (average 1980–89) and under climate change scenarios; SSP2–4.5 and SSP5–8.5 in the 2050s (average between 2050 and 2059s) and 2059s) and 2099s (average between 2090 and 2099) from May to October. A and C) For Embryos, Yolk-sac larvae and feeding larvae in Vanderhoof Braided Section for the period when the life stages are present in the Nechako River. B and D) For Juveniles in other habitats from May to October. The long dashed line indicated a thermal exposure risk of 1. Error bars represent SD.



Figure 6: Thermal exposure risk spatial map for white sturgeon of the Nechako River life stages; embryo, yolk-sac larvae and larvae under SSP2–4.5 and SSP5–8.5 in the 2050s (average 2050–59). (A, B) For the embryo life stage occurring between 15 May and 24 June. (C, D) For the yolk-sac larvae life stage occurring between 24 May and 7 July. (E, F, G, H) For the larvae life stage occurring between 20 June and 31 July. Cool and warm colours represent low and high thermal exposure risk, respectively.



Figure 7: Thermal exposure risk spatial map for white sturgeon of the Nechako River life stages; juvenile under SSP2-4.5 and SSP5-8.5 by the 2050s (average 2050-59) and the 2090s (average 2090-99) for the STMP months (July and August). (A, B) Map for July and August under SSP2-4.5 by the 2050s. (C, D) Map for July and August under SSP5-8.5 by the 2050s. (E, F) Map for July and August under SSP2-4.5 by the 2090s. (G, H) Map for July and August under SSP5–8.5 by the 2090s. Cool and warm colours represent low and high thermal exposure risk, respectively.

to water temperatures exceeding the 18°C threshold for the period, exposing other species to stress.

sockeye salmon (Macdonald et al., 2012). This could lead juveniles or warmer conditions outside the STMP's active

The analysis focused on specific locations rather than the whole river provided informative contrast. Habitat-specific evaluation in the 1980s, such as the Vanderhoof Reach, showed T_e values <1 for the embryo, yolk-sac larvae and feeding larvae life stages. The Vanderhoof Reach has been identified as an important habitat for white sturgeon conservation in the Nechako River (SARA, 2002; Fisheries and Canada, 2014). However, except for Fraser Lake in July, all juvenile critical habitats showed T_e values >1. Although remediating suitable substrates within key spawning habitats is essential to restoring recruitment (McAdam et al., 2018), our results suggest that climate-related increases in water temperature could limit future recovery. Previous studies have noted the impact of thermal stress on the survival and growth of white sturgeon (Wang et al., 1985; Boucher et al., 2014; Earhart et al., 2023). As such, the life stage species analysis in our study highlights stage-specific risks, which can guide both research and future mitigation.

Our study highlights that the embryonic and yolk-sac larvae life stages are less vulnerable to thermal risk than the feeding larvae (both feeding larvae18 and feeding larvae20) due to the differences in the time of year when these life stages are present. This finding is significant, as these earlier stages coincide with critical periods when substrate changes are associated with recruitment failure (McAdam, 2015; McAdam et al., 2005). In the 2050s, the embryos and yolk-sac larvae are projected to experience a thermal risk of <1, regardless of climate change scenario. However, by the 2090s under the high-emission scenario (SSP5-8.5), the risk is projected to exceed 1 compared to the 1980s (Fig. 5). Such high thermal risk increases have significant impacts on the spawning, incubation and survival of white sturgeon (Wang et al., 1985; Deng et al., 2002), as increasing temperatures are linked to increased mortality rates and physical abnormalities (Van Eenennaam et al., 2005; Leal et al., 2021). These results show that the embryos and yolk-sac larvae display resilience, which provides valuable guidance for the timing of future mitigation. Feeding larvae will face high exposure across various climatic scenarios and time frames, especially during the warmer summer months when this life stage occurs. More studies are needed to understand how sturgeon larvae respond to changing conditions, especially given the lack of thermal tolerance studies for this life stage. Our model outcomes suggest that investigation of both peak temperatures and longer duration exposures would be informative.

Rising thermal risk for the juvenile stage emphasizes the need for improved thermal mitigation in regulated rivers (Cheng *et al.*, 2020). While the current STMP offers some protection during the summer for vulnerable migratory sockeye salmon, it may not adequately address the thermal impacts of climate change on multiple species. The ability for upstream water releases to achieve specific downstream mitigation targets (e.g. 20°C) is expected to diminish as rising temperatures increase reservoir outflow temperatures over the coming decades (Fig. 3). Such challenges are already becoming apparent because of the effects of drought in 2023 and

.....

2024 (CBC News, 2024), as limitations on reservoir refilling under sequential years of drought will limit the ability to provide summer cooling flows if drought conditions continue. While other recovery measures aim to mitigate the causes of recruitment collapse, our findings indicated that climate warming poses significant challenges for future species recovery. Further research is needed to refine thermal tolerance for various life stages and understand the timeline of these risks.

Study limitations

Our methodology integrates a spatially distributed hydrological model with the physiological tolerance limits of white sturgeon's early life stages under climate change scenarios. This framework allows us to assess climate-induced impacts on white sturgeon in the Nechako River, enabling a detailed examination of the thermal vulnerability across specific river reaches relevant to different life stages. Nonetheless, it is important to acknowledge the inherent limitations of our approach.

Our water temperature prediction model may face accuracy issues due to uncertainties in the input meteorological and observed water temperature data used for calibration (Khorsandi et al., 2022; Yoshida et al., 2022). The model's precision is influenced by the resolution of physical catchment properties and the aggregation or disaggregation process (Markhali et al., 2022). While there is no significant systematic bias in our simulations, the CEQUEAU model tends to underestimate the extreme temperatures, which could downplay impacts. The biases are linked to some potentially faulty calibration data and the presence of large water bodies (Khorsandi et al., 2023). As ecosystem health deteriorates and weather systems become more unstable, the predictive accuracy of current models may decline. This highlights the importance of regular validation and updates to account for emerging patterns. Careful consideration of the data preparation step is essential to address these challenges and improve the accuracy of the model.

Predicting the optimal temperature and T_e for white sturgeon in the Nechako River based on laboratory data has limitations due to the complexity of ecosystems and speciesenvironment interactions. For instance, the optimal temperature range of 14 and 18°C (Table 1) contrasts with the broader natural spawning range of 11-18°C (Sykes and Bio, 2010). Additionally, juvenile survival remains high even when summer temperatures exceed a T_e threshold of 18°C (Buckner et al., 2024). These discrepancies highlight the necessity for a more ecologically inclusive approach to evaluate thermal limits and exposure risks for white sturgeon. Relying on data from short-term laboratory studies data does highlight areas of concern but may also overlook factors that influence responses in the natural environment, including acclimation and adaptation to elevated temperatures, potentially leading to misinformed predictions and management decisions. Natural thermal variations should be considered when evaluating the physiological performance of wild species (Morash *et al.*, 2018).

Our framework's inability to account for the frequency and duration of juvenile and adult movements, such as seeking cooler areas or deeper waters, limits its effectiveness. This may underestimate the species' resilience and adaptive strategies, reducing its applicability for real-time conservation and management decisions.

Conclusion

Our study highlights the significant impact of climate change and water management practises on the optimal thermal performance and survival of white sturgeon in the Nechako River, British Columbia. Using the CEQUEAU hydrologicalthermal model, we projected future water temperatures and assessed the T_e for early life stages of white sturgeon under two climate scenarios (SSP2-4.5 and SSP5-8.5). A Te value of 1 signifies the assumed optimal temperature for white sturgeon, where their performance is at its peak. When T_e values exceed 1, it denotes a higher risk of thermal exposure, with higher T_e values suggesting more severe stress and reduced performance. Embryos and yolk-sac larvae showed resilience to water temperature increases whereas feeding larvae and juveniles experienced significantly higher Te values, often exceeding 1. Modelling outcomes show that current water management under the STMP will not adequately address the thermal requirements of white sturgeon under projected future climate conditions. Increasing T_e values for feeding larvae and juveniles highlight the need to include white sturgeon in future thermal mitigation strategies. Overall, our study provides valuable insights into the complex relationship between water management, climate change and the thermal ecology of white sturgeon in the Nechako River, emphasizing the need for proactive and adaptive conservation measures.

Author contributions

M.A.O.: conceptualization, methodology, validation, visualization, formal analysis, data curation, writing—original draft, writing—review and editing; M.K.: methodology, data curation, writing—review and editing; R.P.: methodology, writing—review and editing; M.L.E.: methodology, writing review and editing; R.A.: writing—review and editing; S.M.A.: validation, writing—review and editing; C.J.B.: validation, writing—review and editing; A.S.-H: conceptualization, validation, funding acquisition, writing—review and editing.

Conflicts of interest

The authors have declared that no competing interests exist.

.....

Funding

This work was funded by the Canadian Natural Sciences and Engineering Research Council 17 (NSERC) and Rio Tinto as part of a Collaborative Research and Development grant (Grant 18 Number: CRDPJ 523640-18).

Data availability

The data that supports the findings of this study are openly available at https://doi.org/10.5683/SP3/EK7Y7Y.

Supplementary material

Supplementary Material is available at *Conservation Physiology* online.

References

- Arsenault R, Poulin A, Côté P, Brissette F (2014) Comparison of stochastic optimization algorithms in hydrological model calibration. *Journal* of Hydrologic Engineering 19: 1374–1384. https://doi.org/10.1061/ (ASCE)HE.1943–5584.0000938.
- Auffray M, Senécal J-F, Turgeon K, St-Hilaire A, Maheu A (2023) Reservoirs regulated by small dams have a similar warming effect than lakes on the summer thermal regime of streams. *Sci Total Environ* 869: 161445. https://doi.org/10.1016/j.scitotenv.2023.161445.
- Bernhardt ES, Savoy P, Vlah MJ, Appling AP, Koenig LE, Hall RO Jr, Arroita M, Blaszczak JR, Carter AM, Cohen M *et al.* (2022) Light and flow regimes regulate the metabolism of rivers. *Proc Natl Acad Sci* 119: e2121976119. https://doi.org/10.1073/pnas.2121976119.
- Boucher M, McAdam S, Shrimpton J (2014) The effect of temperature and substrate on the growth, development and survival of larval white sturgeon. *Aquaculture* 430: 139–148. https://doi.org/10.1016/ j.aquaculture.2014.03.011.
- Buckner JH, Davies T, McAdam DSO, Taylor EB, Waples RS, Baskett ML (2024) Long life spans can mitigate the genetic effects of strays from temporary conservation hatchery programs. *Canadian Journal of Fisheries and Aquatic Sciences, ja* 82: 1–15. https://doi.org/10.1139/ cjfas-2023–0355.
- Bush E, Bonsal B, Derksen C, Flato G, Fyfe J, Gillett N, Greenan BJW, James TS, Kirchmeier-Young M, Mudryk L et al. (2022) Canada's Changing Climate Report in Light of the Latest Global Science Assessment. Government of Canada. Ottawa, ON. 37p.
- Carron JC, Rajaram H (2001) Impact of variable reservoir releases on management of downstream water temperatures. *Water Resour Res* 37: 1733–1743. https://doi.org/10.1029/2000WR900390.
- CBC News. (2024). 'Every drop counts'. B.C. prepares for impacts of ongoing drought; Province is entering 'unfamiliar territory,' fore-cast centre warns. CBC News. 2024 Brenna Owen. The Canadian Press. https://www.cbc.ca/news/canada/british-columbia/drought-unfamiliar-territory-1.7199512

Cheung, K. (2019) The effects of embryonic incubation temperature on subsequent development, growth, and thermal tolerance through early ontogeny of white sturgeon (T). University of British Columbia. Retrieved from https://open.library.ubc.ca/collections/ ubctheses/24/items/1.0379560.

.....

- Cheng Y, Voisin N, Yearsley JR, Nijssen B (2020) Thermal extremes in regulated river systems under climate change: an application to the southeastern US rivers. *Environ Res Lett* 15: 094012. https://doi.org/ 10.1088/1748–9326/ab8f5f.
- Cheung WW, Oyinlola MA (2019) Dynamic Integrated Marine Climate, Biodiversity, Fisheries, Aquaculture and Seafood Market Model (DIVERSE). Faculty Research and Publications. R. Retrieved from https://open.library.ubc.ca/collections/ facultyresearchandpublications/52383/items/1.0387414.
- Clarke A, Pörtner H-O (2010) Temperature, metabolic power and the evolution of endothermy. *Biol Rev* 85: 703–727. https://doi.org/10.1111/ j.1469-185X.2010.00122.x.
- Daigle A, St-Hilaire A, Ouellet V, Corriveau J, Ouarda TB, Bilodeau L (2009) Diagnostic study and modeling of the annual positive water temperature onset. J Hydrol 370: 29–38. https://doi.org/10.1016/j. jhydrol.2009.02.042.
- Deng X, Van Eenennaam JP, Doroshov SI (2002) Comparison of Early Life Stages and Growth of Green and White sturgeon 28237: 248.
- DFO (2014) *Recovery Strategy for White Sturgeon (Acipenser transmontanus) in Canada [Final].* Fisheries and Oceans Canada, In Species at Risk Act Recovery Strategy Series. Ottawa, p. 252.
- Earhart ML, Blanchard TS, Morrison PR, Strowbridge N, Penman RJ, Brauner CJ, Schulte PM, Baker DW (2023) Identification of upper thermal thresholds during development in the endangered Nechako white sturgeon with management implications for a regulated river. *Conservation Phys Ther* 11: coad032. https://doi.org/10.1093/ conphys/coad032.
- Eyring V, Bony S, Meehl GA, Senior SA, Stevens B, Stouffer RJ, Taylor KE (2016) Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development* 9: 1937–1958.. https://doi.org/10.5194/ gmd-9-1937-2016.
- Eliason EJ, Anttila K (2017) Temperature and the cardiovascular system. *Fish Physiology* 36: 235–297. https://doi.org/10.1016/bs. fp.2017.09.003.
- Fernandes JA, Rutterford L, Simpson SD, Butenschön M, Frölicher TL, Yool A, Cheung WW, Grant A (2020) Can we project changes in fish abundance and distribution in response to climate? *Glob Chang Biol* 26: 3891–3905. https://doi.org/10.1111/gcb.15081.
- Fisheries, & Canada, O (2014) Recovery strategy for white sturgeon (Acipenser transmontanus) in Canada [Final]. In *Species at Risk Act Recovery Strategy Series*, Fisheries and Oceans Canada Ottawa, p. 252. https://www.canada.ca/en/environment-climatechange/ services/species-riskpublic-registry/recoverystrategies/whitesturgeonamended.html

- Gatien P, Arsenault R, Martel J-L, Khorsandi M, St-Hilaire A (2024) Evaluating the effects of climate change on river water temperature downstream of a hydropower reservoir in western Canada. *Canadian Water Resources Journal/Revue Canadienne Des Ressources Hydriques* 49: 402–420. https://doi.org/10.1080/07011784.2024.2314098.
- Geist J (2011) Integrative freshwater ecology and biodiversity conservation. *Ecol Indic* 11: 1507–1516. https://doi.org/10.1016/ j.ecolind.2011.04.002.
- Green A, Ettya A, McPhee S, Ricker B, Temenos C (2014) British Columbia in a Global Context. Faculty Research and Publications. B, BCcampus Open Education. Retrieved from https://open. library.ubc.ca/collections/facultyresearchandpublications/52383/ items/1.0397014.
- Guderley H (2004) Metabolic responses to low temperature in fish muscle. *Biol Rev* 79: 409–427. https://doi.org/10.1017/ S1464793103006328.
- Halsey LG, Matthews PGD, Rezende EL, Chauvaud L, Robson AA (2015) The interactions between temperature and activity levels in driving metabolic rate: theory, with empirical validation from contrasting ectotherms. *Oecologia* 177: 1117–1129. https://doi.org/10.1007/ s00442-014-3190-5.
- Hildebrand LR, Drauch Schreier A, Lepla K, McAdam SO, McLellan J, Parsley MJ, Paragamian VL, Young SP (2016) Status of White Sturgeon (Acipenser transmontanusRichardson, 1863) throughout the species range, threats to survival, and prognosis for the future. *Journal* of Applied Ichthyology 32: 261–32. Portico. https://doi.org/10.1111/ jai.13243.
- IPCC. (2023). Summary for Policymakers. In: Climate Change 2023: Synthesis Report. A Report of the Intergovernmental Panel on Climate Change. Contribution of Working Groups I, II and III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Core Writing Team, H. Lee and J. Romero (eds.)]. IPCC, Geneva, Switzerland, (in press).
- Jay KJ, Crossman JA, Scribner KT (2020) Temperature affects transition timing and phenotype between key developmental stages in white sturgeon Acipenser transmontanus yolk-sac larvae. *Environ Biol Fishes* 103: 1149–1162. https://doi.org/10.1007/ s10641-020-01007-1.
- Khorsandi M, St-Hilaire A, Arsenault R (2022) Multisite calibration of a semi-distributed hydrologic and thermal model in a large Canadian watershed. *Hydrol Sci J* 67: 2147–2174. https://doi. org/10.1080/02626667.2022.2132161.
- Khorsandi M, St-Hilaire A, Arsenault R, Martel J-L, Larabi S, Schnorbus M, Zwiers F (2023) Future flow and water temperature scenarios in an impounded drainage basin: implications for summer flow and temperature management downstream of the dam. *Clim Change* 176: 164. https://doi.org/10.1007/s10584-023-03634-w.
- Leal MJ, Van Eenennaam JP, Schreier AD, Todgham AE (2021) Reduced growth may be linked to lower aerobic scope in juvenile triploid white sturgeon (Acipenser transmontanus). *Aquaculture* 534: 736157. https://doi.org/10.1016/j.aquaculture.2020.736157.

- Macdonald J, Morrison J, Patterson D (2012) The efficacy of reservoir flow regulation for cooling migration temperature for sockeye salmon in the Nechako River watershed of British Columbia. *North American Journal of Fisheries Management* 32: 415–427. https://doi.org/10.1080/02755947.2012.675946.
- Markhali SP, Poulin A, Boucher M (2022) Spatio-temporal discretization uncertainty of distributed hydrological models. *Hydrol Process* 36: e14635. https://doi.org/10.1002/hyp.14635.
- McAdam DSO (2012) Diagnosing Causes of White Sturgeon (Acipenser transmontanus) Recruitment Failure and the Importance of Substrate Condition to Yolksac Larvae Survival. Electronic Theses and Dissertations (ETDs). University of British Columbia. Retrieved from https:// open.library.ubc.ca/collections/ubctheses/24/items/1.0073212.
- McAdam DSO (2015) Retrospective weight-of-evidence analysis identifies substrate change as the apparent cause of recruitment failure in the upper Columbia River white sturgeon (Acipenser transmontanus). *Can J Fish Aquat Sci* 72: 1208–1220. https://doi.org/10. 1139/cjfas-2014–0423.
- McAdam S, Crossman J, Williamson C, St-Onge I, Dion R, Manny B, Gessner J (2018) If you build it, will they come? Spawning habitat remediation for sturgeon. J Appl Ichthyol 34: 258–278. https://doi. org/10.1111/jai.13566.
- McAdam SO, Walters CJ, Nistor C (2005) Linkages between white sturgeon recruitment and altered bed substrates in the Nechako River, Canada. *Trans Am Fish Soc* 134: 1448–1456. https://doi.org/10.1577/ T04-199.1.
- Mishra GJ, Kumar AU, Tapas MR, Oggu P, Jayakumar K (2023) Evaluating hydrological alterations and recommending minimum flow release from the Ujjani dam to improve the Bhima River ecosystem health. *Water Science & Technology* 88: 763–777. https://doi.org/10.2166/wst.2023.236.
- Morash AJ, Neufeld C, MacCormack TJ, Currie S (2018) The importance of incorporating natural thermal variation when evaluating physiological performance in wild species. *J Exp Biol* 221: jeb164673. https:// doi.org/10.1242/jeb.164673.
- Morin G, Couillard D (1990) Predicting river temperatures with a hydrological model. *Encyclopedia of Fluid Mechanic, Surface and Groundwater Flow Phenomena* 10: 171–209.
- Morin G, Paquet P (1995) Le modèle de simulation de quantité et de qualité CEQUEAU: guide de l'utilisateur vesion 2.0 pour Windows. INRS-Eau, rapport de recherche no. 435, pp. 54. https://espace.inrs. ca/id/eprint/1099.
- Muñiz-González A-B, Martínez-Guitarte J-L (2020) Combined effects of benzophenone-3 and temperature on gene expression and enzymatic activity in the aquatic larvae Chironomus riparius. *Sci Total Environ* 698: 134292. https://doi.org/10.1016/j.scitotenv.2019.134292.
- Oyinlola, M. A., Khorsandi, M., Mayer, N., Butler, N., Van Wert, J. C., Eliason, E. J., Arsenault, R., Brauner, C. J., Hinch, S. G., & St-Hilaire, A. (2024). *Thermal Exposure Risk in Different Life Stages of Chinook Salmon in the Nechako River system, British Columbia. Climatic Change*, 177: 1–12. https://doi.org/10.1007/s10584-024-03833-z, 1, 25.

- Oyinlola MA, Khorsandi M, Penman R, Earhart ML, Arsenault R, Brauner CJ, St-Hilaire A (2023) Hydrothermal impacts of water release on early life stages of white sturgeon in the Nechako river (BC Canada). *J Therm Biol* 117: 103682. https://doi.org/10.1016/j. jtherbio.2023.103682.
- Palmer MA, Lettenmaier DP, Poff NL, Postel SL, Richter B, Warner R (2009) Climate change and river ecosystems: protection and adaptation options. *Environ Manag* 44: 1053–1068. https://doi.org/10.1007/ s00267-009-9329-1.
- Penman RJ, Bugg W, Rost-Komiya B, Earhart ML, Brauner CJ (2023) Slow heating rates increase thermal tolerance and alter mRNA HSP expression in juvenile white sturgeon (Acipenser transmontanus). *J Therm Biol* 115: 103599. https://doi.org/10.1016/j.jtherbio.2023.103599.
- Perry AL, Low PJ, Ellis JR, Reynolds JD (2005) Climate change and distribution shifts in marine fishes. *Science* 308: 1912–1915. https://doi. org/10.1126/science.1111322.
- Poff N, Brinson MM, Day J (2002) Aquatic ecosystems and global climate change. *Pew Center on Global Climate Change, Arlington, VA* 44: 1–36.
- van Poorten, B. T., McAdam, D. S. O., Williamson, C., Schwarz, C. J., & Carruthers, T. R. (in review). Estimating Spatial Distribution, Population Size and Trajectory of Nechako White Sturgeon Using a Parametrically Concise Multi-State Capture-Recapture Model.
- Rantanen M, Karpechko AY, Lipponen A, Nordling K, Hyvärinen O, Ruosteenoja K, Vihma T, Laaksonen A (2022) The Arctic has warmed nearly four times faster than the globe since 1979. *Communications Earth & Environment* 3: 168. https://doi.org/10.1038/ s43247-022-00498-3.
- Reid AJ, Carlson AK, Creed IF, Eliason EJ, Gell PA, Johnson PT, Kidd KA, MacCormack TJ, Olden JD, Ormerod SJ *et al.* (2019) Emerging threats and persistent conservation challenges for freshwater biodiversity. *Biol Rev* 94: 849–873. https://doi.org/10.1111/brv.12480.
- Riahi K, Van Vuuren DP, Kriegler E, Edmonds J, O'neill BC, Fujimori S, Bauer N, Calvin K, Dellink R, & Fricko O. (2017) The shared socioeconomic pathways and their energy, land use, and greenhouse gas emissions implications: an overview. *Glob Environ Chang* 42: 153–168. https://doi.org/10.1016/j.gloenvcha.2016.05.009.
- Richter BD, Mathews R, Harrison DL, Wigington R (2003) Ecologically sustainable water management: managing river flows for ecological integrity. *Ecol Appl* 13: 206–224. https://doi.org/10.1890/1051– 0761(2003)013[0206:ESWMMR]2.0.CO;2.
- Rose KC, Bierwagen B, Bridgham SD, Carlisle DM, Hawkins CP, Poff NL, Read JS, Rohr JR, Saros JE, Williamson CE (2023) Indicators of the effects of climate change on freshwater ecosystems. *Clim Change* 176: 23–20. https://doi.org/10.1007/s10584-022-03457-1.
- SARA (2002) Species at Risk Act: An Act Respecting the Protection of Wildlife Species at Risk in Canada. Bill C-5, 1st Session 37th parliament, 49-50-51.

- Sardella BA, Cooper J, Gonzalez RJ, Brauner CJ (2004) The effect of temperature on juvenile Mozambique tilapia hybrids (Oreochromis mossambicus x O. Urolepis hornorum) exposed to full-strength and hypersaline seawater. *Comp Biochem Physiol A Mol Integr Physiol* 137: 621–629. https://doi.org/10.1016/j.cbpb.2003.12.003.
- Schnorbus M, Werner A, Bennett K (2014) Impacts of climate change in three hydrologic regimes in British Columbia. *Canada Hydrological Processes* 28: 1170–1189. https://doi.org/10.1002/hyp.9661.
- Schulte PM (2015) The effects of temperature on aerobic metabolism: towards a mechanistic understanding of the responses of ectotherms to a changing environment. *J Exp Biol* 218: 1856–1866. https://doi.org/10.1242/jeb.118851.
- Song C, Omalley A, Roy SG, Barber BL, Zydlewski J, Mo W (2019) Managing dams for energy and fish tradeoffs: what does a win-win solution take? *Sci Total Environ* 669: 833–843. https://doi.org/10.1016/j. scitotenv.2019.03.042.
- St-Hilaire A, Boucher M-A, Chebana F, Ouellet-Proulx S, Zhou QX, Larabi S, Dugdale S, Latraverse M (2015) Breathing a new life to an older model: the CEQUEAU tool for flow and water temperature simulations and forecasting. Proceedings of the 22nd Canadian Hydrotechnical conference (Vol. 29).
- Sykes G, Bio R (2010) Nechako White Sturgeon Monitoring. Triton Environmental Consultants Ltd. Kamloops, BC. https://

.....

nechakowhitesturgeon.org/uploads/Reports/2009_Spawn_ Monitoring_lowres.pdf.

- Van Eenennaam JP, Linares-Casenave J, Deng X, Doroshov SI (2005) Effect of incubation temperature on green sturgeon embryos, Acipenser medirostris. *Environ Biol Fishes* 72: 145–154. https://doi. org/10.1007/s10641-004-8758-1.
- Wang YL, Buodington RK, Doroshov SI (1987) Influence of temperature on yolk utilization by the white sturgeon, Acipenser transmontanus. *Journal of Fish Biology* 30: 263–271. Portico. https://doi.org/10.1111/ j.1095-8649.1987.tb05751.x.
- Wang YL, Binkowski FP, Doroshov SI (1985) Effect of temperature on early development of white and lake sturgeon, Acipenser transmontanus and A. fulvescens. *Environ Biol Fishes* 14: 43–50. https://doi. org/10.1007/BF00001575.
- Yoshida T, Hanasaki N, Nishina K, Boulange J, Okada M, Troch P (2022) Inference of parameters for a global hydrological model: identifiability and predictive uncertainties of climate-based parameters. *Water Resour Res* 58: e2021WR030660. https://doi.org/10.1029/2021 WR030660.
- Zarri LJ, Danner EM, Daniels ME, Palkovacs EP (2019) Managing hydropower dam releases for water users and imperiled fishes with contrasting thermal habitat requirements. *JAppl Ecol* 56: 2423–2430. https://doi.org/10.1111/1365–2664.13478.